Author’s response to reviews

Title: Effect of Pre-discharge cardiopulmonary fitness on Outcomes in Patients with ST-Elevation Myocardial Infarction after percutaneous coronary intervention

Authors:

He Cai (809051843@qq.com)

Yang Zheng (zhengyanghappy07@sina.com)

Zhaoxi Liu (lzx620727@126.com)

Xinying Zhang (2006zhangxy543@163.com)

Rongyu Li (382272745@qq.com)

Wangshu Shao (shaowangshu96@163.com)

Lin Wang (1051161270@qq.com)

Lin Zou (281642078@qq.com)

Pengyu Cao (caopy1979@outlook.com)

Version: 3 Date: 20 Aug 2019

Author’s response to reviews:

Dear Editor,

Thank you very much for reviewing our manuscript (BCAR-D-19-00237) entitled: “Effect of Pre-discharge cardiopulmonary fitness on Outcomes in Patients with ST-Elevation Myocardial Infarction after percutaneous coronary intervention”. We carefully addressed the comments of the Editor. Please find attached our itemized replies.

Best wishes,

He Cai, MD, PhD

Pengyu Cao, MD, PhD
1. It has come to our attention that throughout your manuscript there is significant overlap with previously published works, in particular:

Discussion:


Reply: Thanks for your comments, we apologized for our negligence, in the new version, we have corrected these overlapping words and sentence.

2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Reply: we have done it in our new version.